Author’s response to reviews

Title: Does air pollution play a role in infertility?: a systematic review

Authors:

Julie Carre (pingou_mouep@hotmail.com)
Nicolas Gatimel (gatimel.n@chu-toulouse.fr)
Jessika Moreau (moreau.je@chu-toulouse.fr)
Jean PARINAUD (parinaud.j@chu-toulouse.fr)
Roger Leandri (leandri.r@chu-toulouse.fr)

Version: 2 Date: 13 Jul 2017

Author’s response to reviews:

We have done all changes suggested in the marked manuscript. You will find the revised files clean and also with the marked modification suggested by you.